Financialisation and economic structure: a two way relationship

Jimena Castillo¹

Abstract

Financialisation is not a homogeneous process. To fully understand its characteristics and implications, it is essential to use an integrated perspective that considers the financial side of the economy and the productive characteristics of countries and the interactions among both. In order to do that, this paper bases its explanation on the ideas of Latin American Structuralism to enrich the understanding of financialisation phenomena in emerging countries. Financialisation does not relieve countries from the centre-periphery dichotomy; on the contrary, it reinforces it. By widening structural heterogeneity, strengthening specialisation in commodities, posing Balance of Payment constraints, limiting the space for industrial and fiscal policies or aggravating financial dependency and cementing the position of peripheral currencies. The analysis emphasises on the channels of transmission and the consequences of financialisation from the perspective of peripheral countries. The paper also claims that financialisation in the periphery takes a specific form according to its structural characteristics; the process is **selective** and **functional**. Selective, as it is focalised in some activities and actors, and functional, for maintaining the position of these countries in the world order, while unaltering internal political balances.

¹ bnmjca@leeds.ac.uk

This is an early draft so please treat it confidentially and do not share, quote, or reference without my knowledge.

1. Introduction

The world economy is characterised by a disproportionate increase in the financial sector vis-à-vis the real economy, extraordinary growth in financial wealth and financial innovation, and the FIRE sector (finance, insurance, real estate). That characterises the US and most of the OECD countries; the changes have been called under the word 'financialisation'. After the economic crisis in the developed world of 2009, there is a thriving interest in the phenomenon across social sciences, particularly among heterodox economists. Still, what financialisation means for developing countries is less studied, but there is a consensus that financialisation in those countries presents distinctive characteristics than in rich ones (Bonizzi, 2013; Karwowski, 2020).

Some authors have indicated that there are varieties of financialisation (Lapavitsas and Powell 2013, Brown et al. 2017). Part of the differences are based on commonalities among these countries institutions and integration into the global financial system. The differences are based on macroeconomic aspects (Engelen et al. 2010, Lapavitsas 2013, Lapavitsas and Powell 2013, Karwowski and Stockhammer 2017, Karwowski et al. 2019). Part is related to the fact that financialisation is externally driven (Lapavitsas 2009, Stockhammer, 2010), imposed by imperialists relations, characterised as a subordinated type of financialisation (Powell 2013, Powell et al. 2020). Measures were imposed favouring the interests of wealthy nations. As a result of capital account liberalisation, negative consequences such as high instability and economic crises were experienced (Dymski 1999, Arestis and Glickman 2002). Latin America acted as an experimental area for deregulation of capital accounts and financial rules, where policies were conducted under authoritarian regimes in the 1970s (for an analysis of the role played by International Financial Institutions and actors, see Déniz and Marshall, 2018).

Another crucial feature that shapes financialisation and the insertion in the international financial system is the position of a country in the currency hierarchy. Currencies from ECs (emerging countries) are located in the lower part of the pyramid, based on the inability to borrow in their own currency, so they need to

engage in operations in US dollars, posing constraints on the balance of payment and economic growth (Becker et al., 2010; Lapavitsas and Powell, 2013; Kaltenbrunner and Painceria, 2015; de Paula et al. 2017). The impossibility of currencies from ECs to fulfil the traditional functions of money need to be compensated by high levels of profitability, which can be evident in the interest rate differentials to attract capital flows (Vernengo, 2005; Kaltenbrunner 2015, da Paula, Fritz, Prates 2017, Bonizzi and Kaltenbrunner 2019).

These explanations primarily focus on the financial and monetary specificities of EC's financialisation, overlooking the connection with the productive side, missing in that way that the integration of ECs into the global productive system also takes a subordinate role. Those who included the productive subordination looked at it from the US perspective without analysing what this meant for financialisation in ECs (Milberg, 2008). On the other hand, among the studies that incorporate some form of productive subordination (Powell, 2013) did it with a focus on those firms that internationalized by integrating into Global Productive Networks (GPN) (Bonizzi et al. 2020), but did not focus on how this implied changes for the entire countrie's productive structure, including those firms that are out of the international productive circuits. There are no studies that investigate financialisation with a focus on the effects in the productive structure, for example differentiating the economic sector affected, the consequences on value-added activities (high vs low value-added) or the impact on segmented labour markets. They usually treat ECs as homologous to central countries, missing in that way one crucial feature: the effects are not homogeneous.

The 'theory' adopted here indicates that financialisation in ECs must be understood in relation to the international insertion in the world economy, taking into consideration the productive dimension. Therefore, it is necessary to make the financial sphere and the structural characteristics of ECs dialogue. To construct an approach that analyses the interdependencies of finance and production, this paper adopts from Latin American Structuralists' writings concepts about technology, external constraints, and the macroeconomic policies followed to cope with business and financial cycles (Prebisch, 1949; Tavares and Gomes, 1998)².

_

² In Pérez Caldentey (2015), there is a detailed analysis of Structuralism's contributions in different periods by author.

According to the 1949 'manifesto'³ of Prebisch, the level of development between countries, those classified as 'central' and 'peripheral', showed the characteristics of their respective economic structures. The theory's emphasis was given to the international insertion in the world; the periphery specialised in primary commodity production. The gains from trade were not equally distributed among both poles, rejecting the Ricardian comparative advantage discourses (Rodriguez, 2001). Commercial, technological, and financial dependency were critical aspects in explaining the conditions of Latin American countries, limiting development in the region (Furtado, 1973; 1984).

In terms of the productive dimension, and concerning technological capabilities and capital accumulation the world today can be interpreted as a two poles system (Ocampo, 2014). Technology is generated mainly in rich countries (Arocena and Sutz, 2000). Those growing faster include technological content in their exporting baskets where the ones relying on natural resources grow more slowly (Ocampo et al. 2009 and Hausmann, 2011). Intellectual property has become a key element, posing difficulties for countries willing to climb up the technological ladder (Cimoli et al., 2005). Latin America has a techno-productive lag, that it is reflected in the participation of engineering-intensive industries in the total industry, patents per habitant, or investment in R&D (Research and development) in terms of GDP (Gross Domestic Product). This is also true, when compared with emerging Asian countries and even with other countries that base their economies on natural resources (ECLAC, 2015 pp. 100). That is why authors as Andreoni and Chang (2021) consider, production is vital for development. In particular the type of production that enables structural change and economic diversification (see Ray, 2010). Nevertheless, we stay here that in order for a country to upgrade technologically, it is necessary to look at the productive side but also consider financialisation and the functioning of the international financial system.

³ Prebisch published the document "The economic development of Latin America and its principal problems" that became to be known as the "ECLAC Manifesto".

Financialisation is a new way of dependency and it is interlinked with the productive sphere, it has different implications for the centre and the periphery; and it needs to be analysed jointly. This paper combines two dimensions to analyse EMs countries position in the world economy (the productive and financial dimensions), trying to answer (a) how financialisation affects those countries' structural characteristics (in particular the productive structure), potentially perpetuating their subordinate position in the global productive system, and, more important how (b) the specific structural characteristics of Latin American economies matters for financialisation phenomena and the insertion in the global financial system. The analysis also considers the domestic institutional drivers and external ones related to economic agents' financial and productive behaviour.

When we refer to the productive structure we are referring to the composition of the output of a country in terms of the material goods and also services produced.⁴ And the structural characteristics are related to its productive structure and those related with the type of integration into the monetary and financial system.

The rest of the paper is organised as follows: Section 2 includes an introduction to the financialisation literature. Section 3 contains the main ideas from Structuralists' writings that are relevant to analyse financialisation phenomena. Section 4 analyses financialisation from a Structuralist point of view. Section 5 concludes.

_

⁴ The productive structure generally refers to the 'the composition of the output of material goods. The economic structure includes, in addition, the production of various kinds of services, including public goods and services' (Hounie et al., 1999 pp 9). In order to simplify we will include in productive structure goods and services.

2. Financialisation

The use of the concept "financialisation" has increased over time, where most of the literature on the topic is produced by heterodox economists (Marxists, Post Keynesian and Regulationists)⁵. Here we follow the definitions from Epstein, who defines financialisation as the "increasing role of financial motives, financial markets, financial actors and financial institutions in the operation of the domestic and international economies" (Epstein, 2005, p. 3). Literature reviews on the topic concluded that the term had been used in several ways (Lapavitsas, 2014; Sawyer, 2014; Van der Zwan, 2014; Epstein, 2015). The research around financialisation is enormous; for the purposes of this paper, the focus will be on those aspects that relate directly or indirectly with the centre-periphery dichotomy and with the productive structure. For Lapavitsas (2014), financialisation is the result of profound changes in capitalist economies reflected on the transformation of non-financial corporations (NFCs), banks, and households. Here, the focus will be on the first two. Households will not be explicitly addressed unless they could impact the structural characteristics of the economy.

The massive increase in the financial sector came with a diversification of its activities and products with more influence globally (Boyer, 2000; Aglietta and Breton, 2001; Duménil and Lévy, 2004; Crotty, 2003; Stockhammer, 2004). The growth has been faster than the real economy, where financial profits to total profits increased (Stockhammer, 2010). The change is represented by the expansion of market capitalisation, turnover, and soaring debt levels in firms, households, and even in the financial sector. Besides, a mark of the new financialised system is the rise of new financial institutions (shadow banks), pension funds, private equity, and hedge funds (Adrian and Shin, 2010; Adrian *et al.*, 2010). Researchers identify a change in banks' business model, shifting away the lending focus from NFCs to households (Lapavitsas, 2009a). Also, an important part of their income comes from fees and commissions (Erturk and Solari, 2007; dos Santos, 2009).

⁵ Nowadays, financialisation is also present in mainstream publications (Englen, 2008).

Financialisation is global in its nature (Christophers, 2012). The new landscape results from active policies on an international scale: deregulation of the financial sector and capital flows, including the lift of interest rates ceilings. Consequently, there is a documented increase across the globe of external assets and liabilities (e.g., Stockhammer, 2010; Akyüz, 2014; Karwowski and Stockhammer, 2017). In Latin America, there is an increase in foreign inflows, outward and inward Foreign Direct Investment (FDI) (UNCTAD, 2014) and a rise in the search for short term capital gains. For Lapavitsas (2014), the international dimension of financialisation has a more substantial influence in developing countries, as these economies are more vulnerable to external conditions. In Latin America, foreign banks were promoted by Washington Consensus reforms, based on the idea that through competition, national institutions would increase their efficiency (Correa, 2006). As dos Santos (2013) argues, foreign banks are crucial in bringing into national economies financialised practices of rich countries, like aggressive strategies towards household lending. There is documented evidence of these practices in peripheral countries (Lapavitsas, 2009a; Gabor, 2012; dos Santos, 2013). The presence of foreign banks did not come along with a massive deepen in the financial system (Manuelito and Jimenez, 2010; de la Torre et al. 2012). Nevertheless, it is essential to analyse financialisation as a process, looking at the level and trajectory (Fernandez and Aalbers, 2020).

In addition, households resorted to financial markets to finance the provision of public goods and services due to the retraction of welfare states (Streeck, 2017). The reorientation of the business strategies of banks fuelled household debt (among others Lapavitsas, 2009a, Gabor, 2010 and 2012; Ashman *et al.*, 2011; dos Santos, 2013; Karacimen, 2016; Karwowski and Stockhammer, 2017) and real estate booms (Aalbers, 2008; Robertson, 2016). Also, households are linked with financial markets by the asset side of their balance sheets, by pensions and insurances (Lapavitsas 2009b); in Latin America, Chile is an extreme example of it.

Financialisation reflects not only the transformations within the financial sector but also in relation to others (Stockhammer, 2004). In the case of NFCs, some of the changes in the behaviour had been called under the term 'shareholder value orientation'; those changes are related to the reward structure of top managers, changing priorities, and focusing on short-term strategies and profitability levels. That

implied a closer relation of NFCs with the financial sector on a national and international scale (Lazonick and O'Sullivan, 2000; Crotty, 2003; Stockhammer, 2004; Hein, 2015). Boyer (2000) named the regime a 'finance-led accumulation regime', characterised by changes in investment behaviour with a redistribution detrimental to labour favouring shareholders.

The international dimension of financialisation is the most important when analysing financialisation in EEs. The surge in cross border flows (Stockhammer 2010) induced by external forces (Lapavitsas 2009) imply massive impacts for domestic economies. The understanding of EE financialisation is called under the concept of subordinated financialisation (Bonizzi et al. 2020, Kaltenbrunner and Painceira 2018, Alami 2019). The change do not only reflect an increase in the quantity of this flows, it is also qualitative, with different players integrated into international financial markets inducing financial innovation (Bortz and Kalternbrunner, 2017). One aspect that is a landmark of this period is that the size of institutional investors are so huge in comparison of a country's economy.

Movements in capital flows are more dependent on external conditions than on the fundamentals of peripheral economies, high liquidity and low-interest rates in the rich world made enormous amounts of capital available in ECs, attracted by higher domestic interest rates. ECs have to offer high-interest rates to compensate for low liquidity and high-risk environments (Paula et al., 2017; Fritz et al., 2018). When an event ends this optimistic trend, the cycle reverses, uncertainty increases, and there is a 'flight to quality' movement where capitals leave peripheral countries. Investors usually migrate to positions in currencies that offer greater liquidity and stability. Uncertainty generates a deceleration of economic growth, ending usually in economic crisis, and 'sudden stops' end in mayor depreciations.

With free capital mobility, these countries are constraint by the decisions of central countries, so even when displaying a free-floating exchange regime, countries cannot apply monetarily policies as they are dependent on the risk perceived by global investors. Small developing economies face a macroeconomic dilemma, not trilemma⁶

⁶ The case of the trilemma, states that under free capital mobility, countries can apply independent monetary policies if they leave the exchange rate to float.

(Rey, 2013). In peripheral economies, central banks do not allow completely free-floating exchange rates, they intervene, but without setting a fixed rate; this is called 'dirty floating.' In that way, they try to stabilize exchange rates (interventions are not free from pressures from national actors, like exporters).

In the last years derived from QE (Quantitative Easing) programs, high availability of cheap capital, flew to ECs buying assets, sovereign bonds and currencies (Bonizzi and Toporowski 2017). Due to inflation targeting regimes in the region with opened capital accounts and floating exchange rates (Frenkel, 2006), central banks needed to engage in sterilisation strategies to control inflation, increasing the offer on public debt securities. These instruments were bought by banks and investment funds, expanding in that way their balance sheets (Kalternbrunner and Painceira, 2017). Aggressive strategies towards household lending implied more household financialisation and a hike in asset prices; this dynamic resulted in more financial fragility (a la Minsky, 1978). In addition, central banks accumulated reserves to cope with changes in financial cycles and liquidity cycles in the centre.

In order to build up the understanding of financialisation in the context of peripheral countries, the following section states the most relevant aspects of Structuralism. This theory indicates that changes in the capitalist economy must be analysed in the context of the centre-periphery dichotomy, as economic phenomena can entail profound differences among them. We understand that financialisation is one of these changes.

3. Structuralism: characteristics of peripheral economies

Latin American structuralism is the most outstanding theoretical contribution to economics and social sciences from the global south (Rodriguez, 2001). According to the international division of labour, the theory reflected that the world is divided into two poles: the centre and the periphery. Both possess distinctive characteristics regarding capital accumulation and technological progress, where the periphery is dependent on the technology that is originated in the centres. The periphery is heterogeneous because labour productivity is high in the exporting sector; meanwhile, the rest of the economic sectors are stagnant, and the economy specialises in few commodity goods. In the centre's case, it presents high levels of labour productivity across economic sectors and produces a wide range of goods with inter-sectorial complementation and integration across economic activities (Rodriguez, 1980). Latin America, in this classification, is a peripheral region. Economic relations between both poles tend to reproduce the conditions of underdevelopment in the periphery and increase the distance between them because central countries appropriate most of the fruits of technical progress.

According to ECLA, the pattern of specialisation is the cause for external imbalances (Hounie et al., 1999). Related to it comes a second key concept of the theory, the secular decline in the terms of trade. The Prebisch-Singer hypothesis indicated that terms of trade between the price of products exported by the centre (industrialised products) and the periphery (commodities) tend to decline over time. They challenged the idea that technological progress will lower prices in the industrial sector in relation to commodities. The secular tendency of the terms of trade to deteriorate is based on arguments related to commodities' characteristics and countries' characteristics (Erten, 2011).

Arguments around commodities include Engel's law, which establishes that the demand for food increases in a smaller proportion when income increases. So, if income rises at a world level, food producer countries will face a fall in demand in relative terms. Also, the price elasticity of demand for commodity products is smaller.

As income expands, the demand for commodities grows less than in the case of industrial goods and, the relative price is less favourable for these goods. Then, peripheral countries need to export a growing amount of goods to keep up with the increased value of industrial exports (Rodriguez, 1980). Besides, commodity products can be substituted by synthetic products, competing for demand.

The rest of the arguments are related to the country's conditions, which are different in the centre and periphery. According to Prebisch, there is surplus labour in peripheral labour markets because of the difficulties of industrialising and migrating to industrial countries. Trade unions are weaker than in central countries. In contrast, in rich countries, powerful unions and no surplus labour guarantees that prices are set with a mark-up on costs. Consequently, wages in the periphery grow less concomitant with price increases. Also, rich countries tend to protect their primary domestic markets, shrinking market sizes. Lastly, the necessity of peripheral countries of importing goods to industrialise makes the industrialisation process more complex. It results in a rise in the price of capital goods when many countries want to buy these goods. The process is exacerbated by international agencies promoting exports without coordination (Singer 1998 in Erten, 2011). That is why commodity production is related to underdevelopment. Erten (2011) indicates that Singer in the 1970s called for a change in the focus of the debate related to the terms of trade, from products to countries. That was because even commodity-producing countries tend to increase the amount of processed goods they trade, at a country level, the evolution in developing countries' trade terms vs. developed ones can still be detrimental for the former ones.

Another crucial aspect introduced by Structuralists is the role of economic cycles. In opposition to Neoclassic economics that focused on equilibrium and the transition to equilibrium, for Prebisch (1949), cycles are inherent to capitalist economies. They have an essential function in the capitalist system. Cycles have different manifestations in both poles; they are originated in the centre and translated into the periphery and are the driver of economic growth in connection with productivity increases. In the upper part of the cycle, the rise in demand for goods in the centres cannot be satisfied by supply, so profits start to increase, and at some point, they generate a rise in prices, correcting in that way the excessive demand.

The rise in profits in the centre is translated into higher demand for peripheral products. This demand impacts commodity prices in the periphery; the magnitude will depend on the time-response of production and stock availability. Overall, commodity prices tend to increase more than industrial ones. Moreover, once exports increase, domestic activity starts to grow and employment and income levels. With the rise in income levels, imports start to increase. In the descending phase, the opposite occurs, the fall in exports lowers income and employment, with the consequent decrease in imports. Besides, commodity prices decline more than industrial ones. The explanation for these price movements (commodities vs industrial prices) is related to the flexibility in both economic systems. In the centres during the upward phase, firms compete for workers, and unions push for increasing wages. But, during the downward phase, there are rigidities for the decrease in wages. However, this does not hold in the periphery, as working forces in primary production have not the same levels of organisation as in industrial countries, so the effects are not symmetric. The asymmetry is related to the higher union organisation of manufacturer workers and related to industrial markets' greater power in setting prices. As Prebisch puts it, the smaller the compression in income in the centre (profits or wages), the more significant it must be in the periphery. That is crucial because industrial centres can then retain the fruits of their technical innovations, and, they can retain part of the technical progress of the periphery.

The main policy recommendations made by ECLA in the 1950s is state-led industrialisation. The process of industrialisation did not disregard the upgrade of technological content of primary production. On the contrary, it was essential to raise living standards and to incorporate equipment, machinery, and instruments to take advantage of technology. Besides, the periphery needed to import capital goods, so any improvement in the exporting sector was crucial for obtaining US dollars. A long-term strategy for development must consider economic cycles as peripheral incipient industries are vulnerable to fluctuations and contingencies in the centres. Therefore, anti-cyclical and industrial policies need to be implemented together. However, these policies are challenging to implement as the necessities for infrastructure are particularly important, and the political costs are high.

Nevertheless, industrialisation must be conducted in a way that can counteract its own limits. If industrial growth implies a significant increase in consumption or an early decrease in the productive effort, that can hinder the social purpose of industrialisation. Therefore, it was important to adapt the pace of imports to the payment capacity given by exports. That was a crucial aspect as Latin American industrial strategy depends on the decision of other countries to buy their products. According to his analysis, in most Latin American countries, spontaneous savings were insufficient to cover their most urgent capital needs so that foreign investments could contribute to increasing productivity per worker.

Seminal works from Prebisch about business cycles in the periphery (1921) stressed that they were connected with the place in the international monetary system and were caused by an unstable international capital market. Speculative bubbles resulted from an asymmetric monetary system, where developed countries impose conditions for the rest of the countries (Lampa, 2021). When working at the Central Bank of Argentina, he pushed for contra-cyclical monetary policies to reduce external volatility. According to him, if risk aversion and interest rates were low in the centres, international loans triggered monetary expansion and a hike in imports in the periphery, worsening the Balance of Payments and pushing for extra loans. In addition, domestic banks expanded the capital spent in speculative activities, and with that, financial fragility increased. When something initiated a reverse in capital flows, that meant a crisis for the periphery. He aimed at imposing capital controls and increasing the level of reserves levels to use in times of capital outflows. He also stressed the necessity of cutting bank credit during the upper phase of the cycle. His observations were the opposite of mainstream economists'; for him, capital volatility in the periphery was a structural and permanent characteristic. Prebisch also prescribed a set of economic reforms to increase the use of local currencies, reduce the dollar dependence based on bilateral payment systems, and create a type of central bank from various nations to be a lender of last resort to decrease the political subordination. This type of measures complemented the industrialisation recommendations focusing on substituting imports to be less dependent on US dollars.

Structuralists ideas highlighted here are the contributions in which this paper will build the nexus with financialisation. The key concepts for the analysis of financialisation from a peripheral point of view are the centre/periphery dichotomy, heterogeneity and the pattern of specialisation, declining terms of trade, and the fact that relations between countries are asymmetrical, where business and financial cycles are originated in the centres.

4. Financialisation and structural characteristics a two-way relationship

Financialisation affects the economic structure of peripheral countries, and it is affected by it. It is a two-way relationship. The economic structure comprehends the productive structure (the composition of the output of a country in terms of the material goods and also services produced) and the financial one (the type of integration into the monetary and financial system). First, we present the effects of financialisation in the economic structural characteristics, it is shown how financialisation can reinforce the subordinated position of peripheral countries (productive and financial). Second, a more novel path, which analyses how the economic structure influences the 'nature of financialisation' of peripheral countries, where the direction goes from the economic structure to financialisation. This section will formulate questions for a further research agenda to comprehend from a Structuralist point of view the financial dynamics in ECs.

4.1 From Financialisation to the structural characteristics

Financialisation influences the economic structure exacerbating the subordinated position of peripheral countries in the global economy. The economic structure characteristics' includes those derived from the integration into global productive system, shaped by the international division of labour (productive structure), and those linked with the integration into the global financial and monetary system (financial structure). We present two main channels through which financialisation

affects the economic structure: the real channel and the financial channel. The real channel includes how financialisation through real investment, commodity prices and Balance of Payment constraints impacts the productive structure, deepening heterogeneity and specialisation in commodities and raw materials. Regarding the financial channel, on the one hand, through the lack of financial resources available for real investment and by limiting the space for industrial and fiscal policies the productive structure is affected; on the other hand, capital mobility and cycles of capital that are originated in the centres and translated into the periphery aggravating financial dependency and cementing a lower position in the currency hierarchy.

4.1.1 Real channel: real investment and R&D

Financialisation can directly impact the productive structure by shaping the behaviour of productive firms concerning investment in capital stocks and R&D and by changing the macroeconomic environment in which they operate. In particular, agendas towards privatisation, the achievement of fiscal balances and the adoption of independent monetary policies while increasing commercial and financial integration are the main features of the macroeconomic environment.

We can distinguish three ways in which finance affects the firm's behaviour: the increase in the tenancy of financial assets and profits derived from financial activities, the shareholder-value orientation, and access to funding. In relation to the first one, the financialisation literature refers to NFCs increasing their activities in financial markets, a marked trend to hold financial assets that generate revenue sometimes more profitable than those obtained with real investment (Krippner, 2005; Orhangazi, 2008). That can imply that financial investment crowds out real investment (crowding-out hypothesis). That is also related to the shareholder value orientation, by which companies are more focused on shareholders' interests' over stakeholders (Van der Zwan, 2014). Managerial changes, led by the shareholder value orientation, tied managers' interests with corporations by paying based on performance. The literature also points towards the transfer of companies' earnings by interest and dividend payments to shareholders and the increase in buyback operations to boost companies share prices (Froud et al., 2000; Lazonick and O'Sullivan, 2000). Leading

companies focus on the extraction of value rather than on its creation, discouraging investment in capital stock or R&D, as the benefits are long-term.

The third is related to the capacity of the firm to access external funds to finance real investment. In the case of ECs, access to finance is crucial for investment, the lack of innovation and investment can be related to problems in capital markets (moral hazard, incomplete information, etc.), and debt can have a positive impact if funds are used in buying assets or investing in intangibles. In addition, the literature on financialisation showed that banks changed their business strategies, moving away from lending to companies, making it more difficult for companies to access funds. Nevertheless, higher levels of indebtedness were used to buy financial assets.

The changes highlighted in the financialisation literature resulted in a decrease in capital stock, employment, and spending in R&D. In developed countries, using country data, Stockhammer (2004) linked financialisation with the slowdown in capital accumulation for the US, UK and France. Also, researchers linked financialisation with a slowdown in real investment at a firm level (Orhangazi, 2008; Onaran and Tori 2016). Also, there is evidence of the same pattern for ECs (Demir, 2007; 2009). Soener (2021), with publicly traded information of NFCs, showed that real capital accumulation has fallen over the past 30 years for the 37 countries in his sample. Nevertheless, little evidence was found that this drop is due to financial accumulation, as the relative value of financial assets and financial income decreased. However, shareholder distributions were boosted over time, particular amongst American corporations. He connects financialisation with the strategy to internationalise by big companies. Rabinovich (2019) found that financial accumulation rates in the US are not as great as previously believed, questioning the crowding-out hypothesis. Nevertheless, he showed how companies use funds in buyback operations.

The innovation processes are also hindered by financialisation; investing in R&D is riskier than investing in capital stocks. The pressure of shareholders in obtaining benefits in the short-term is not compatible with long-term innovation (Lazonick and Mazzucato, 2013; Dosi *et al.*, 2016). Lee *et al.* (2020) found that financialisation affected the quality of innovation for OECD countries, making it less radical but more abundant. Financial markets rewarded companies with more patents, so NFCs focused

on patenting short-term innovation to access funds. For ECs, the story is quite similar. Jibril *et al.* (2018) found that when Brazilian companies relied more on financial assets and financial profits, they invested less in innovation. More generally, Chena *et al.* (2018) showed that the expenditure in R&D declines with financialisation in Latin American countries.

The fourth way in which financialisation affects the productive structure is related to the macroeconomic environment in which firms are operating. Financialisation can lead to more uncertainty, and volatility in financial markets makes it more difficult for firms to invest intangible assets. This type of investment entails risk and needs longer horizons (Stockhammer, 2008). That made firms relocate resources into less productive activities (Arcand et al., 2012). Financial shocks in the short-term can affect the direction of real investment, therefore, the productive structure (Cimoli et al., 2020). In addition, the finance-led accumulation regime relegates the state's role in the productive sphere affecting innovation and investment (Boyer, 2000). In Latin America, due to Washington consensus policies, many indigenous technologies that were created locally in public firms were smashed. Privatisation policies sold firms to private owners, usually foreign, that were not interested in sustaining research and development activities in the region. In addition, a catching-up transition period needs active policy measures that can be opposed by the financial sector (Andreoni and Chang, 2019). We can call this mechanism 'short-term bias'; this can configure a fifth way in which financialisation affects the productive structure.

Previous studies of financialisation assessed the impact of financialisation in ECs as if they were similar to those in wealthy nations. The fact that finance can affect a firm's behaviour by crowding out real investment and re-orienting resources to satisfy shareholders or by the capacity to access funding for long-term investment can lead to different effects in the centre and in the periphery. As discussed in Section 3, structural heterogeneity indicates that there is a variation in productivity between firms; productivity is high in the periphery in the exporting sector (commodity exporters) while low on others. If the decrease in investment is concentrated on sectors with low productivity but trying to improve and catch up with their counterparts, financialisation can hinder this path. We argue here that it is crucial to assess the 'composition effect'. The overall effect of financialisation in a peripheral country will

depend on the sector or activity that is affected, if the activities are technology-intensive and add value to their products and in which sector is located (primary or manufacturing, for example). If financialisation is linked with deindustrialisation strategies, this could pose additional risks for peripheral countries. From Structuralist and Neo structuralist writings, the role of industrialisation and the value-adding process is vital. Countries that industrialised first developed technologies and acquired dominant positions in manufacturing production. So, if financialisation impacts technological capabilities, it affects industrialisation as well.

Moreover, countries experiencing premature deindustrialisation suffer the risk of experiencing vicious cycles of underdevelopment (Andreoni and Chang, 2019). Premature deindustrialisation refers to the fact that developing countries are transitioning to a service type of economy without having gone through a complete industrialisation process (Rodrik, 2016, pp.2). Manufacturing activities are beneficial for economic growth, as it is a technologically dynamic industry, showing unconditional worker productivity convergence. Second, historically it absorbed substantial amounts of unskilled labour, distinguishing it from other high-productivity industries like mining or finance. That is very important for the region. The deindustrialisation trend also affected advanced economies through employment loss, but they perform pretty well in terms of output share (when adjusting for prices, income and demographic trends, see Rodrik, 2016). In Latin America, with deindustrialisation, informality has grown, and productivity has suffered.

Exporting companies or public companies, in general, have more sophisticated structures, but small and medium-sized manufacturing firms, which are dominant in the economic landscape, present lower levels of productivity and a smaller capacity to innovate. If this last type of company absorbs the effect, the difference with the rest can increase. On the contrary, if exporting companies absorb the effect, this could decrease productivity, resulting in a more homogeneous structure. However, on a world scale, this widens the productivity distance with central economies. More research is needed; it matters the size of the impact compared to the rest of the economy and with the centre. If financialisation affects long-term investment and radical innovation, the deterioration in employment conditions or wages in the formal sector can be detrimental to aggregate demand. Therefore, financialisation can be a powerful force preventing structural transformations.

4.1.2 Real channel: terms of trade and balance of payment constrains

Unlike industrialised countries, Latin American countries mainly export raw materials or commodities7. Financialisation affects commodity prices, resulting in changes in resource allocation and productive strategies. From a structuralist point of view, periods of economic expansion in the core are followed by higher demand and commodities prices, and commodity prices tend to move more abruptly in the upswing and downswing than industrial ones (Prebisch, 1949). Nowadays, commodity market movements are derived from supply and demand-side reasons and related to financial motives, affected by the financialisation of commodities and through global financial conditions (Akyuz, 2020). Commodity products were treated as financial products and included in international portfolio investment baskets (Gilbert, 2010; Akyüz, 2012). For example, after the 2000s, institutional investors promoted commodity futures as 'safe' because of the low correlation with stock returns (Tang and Xiong, 2010). They acted as an asset class, like securities, attracting lots of capital and speculation. In addition, indices managed by massive hedge funds, like Black Rock, included stocks and bonds from commodity-producing firms (van Huellen and Abubakar, 2021). Besides, investment banks are linked with derivatives and even invest in physical commodities (Isakson, 2014). Commodity prices showed a boom-and-bust trajectory in the last decade.

Price increases are a powerful motivator to improve output and shift resources to profitable sectors (UNCTAD and FAO, 2017). Commodity prices are also affected by global financial conditions, especially capital flows. Expansionary monetary policies in the centre (low-interest rates and a weak dollar) induce capital flows into the periphery stimulating the surge in commodity prices. Hikes in commodity prices and capital inflows have been moving together on several occasions since the 1970s (Akyuz, 2020). In addition, it is worth mentioning that most countries in the periphery are price takers (ECLA, 2012) (one aspect that is an indicator of productive subordination). Most commodities contracts are settled in US dollars, even when they sell the goods to other peripheral economies (an indicator of monetary subordination, reflecting their place in the currency hierarchy). An increase in commodity prices

⁷ With the exception of Mexico.

induces indebtedness, especially in the sector benefited by the boom, and can help to expand investment. The inflow of money favours producing sectors directly or indirectly, also consumption and higher tax revenues. In addition, the increase in foreign exchange earnings can appreciate the local currency, the real appreciation of the local currency can have an unwanted effect on competitiveness ('Dutch disease'). Deindustrialisation or no growth in the manufacturing sector can be a side effect (Poncela et al., 2017).

Terms of trade, which directly relate to commodity prices (FAO and UNCTAD, 2017), are very relevant to the continent's economic conditions. There is a documented impact on shocks of these variables in economic activity: production, investment, trade balance and in others like inflation and the exchange rate (see Peláez and Sierra, 2016 for a review). Shocks in terms of trade can harm economic performance, and there is evidence that those shocks account for a significant portion of the difference in growth rates between countries (review on FAO and UNCTAD, 2017). Commodity price shocks may also affect debt sustainability, particularly in countries where public finances are reliant on commodity export income.

Related to this specialisation pattern, the Balance of Payment Constraints constitute a significant force that affects Latin American economies. Commodity booms induce GDP growth, but the tension on the balance of payment emerges; consumers want to buy products not produced in Latin American countries, in particular elites (Presbich, 1949). Income elasticity of exports is low compared to imports, so when there is an increase in the GDP in the region, an unbalance emerges; periods of high economic growth come with substantial trade deficits. In the short run, deficits can be financed with capital inflows, but in the long run, the productive structure must change (Bértola, 2015). These countries cannot finance sustained deficits by issuing their own currency; the growth rate must be sustainable with the relationship between export growth and income inelasticity of imports (Thirlwall, 1979).

There is an agreement in the literature that the super boom in commodity prices ended, and the expectation is that it will remain in that way for some years to come (see Akyuz, 2020). So, could it be the case that a decline in commodity prices implies a re-industrialisation strategy in the periphery? That does not seem easy, as industrialisation needs long-term investment and active policies. Also, pro-cyclicality

in credit can make it challenging to obtain funds to finance industrial strategies plus other constraints in policies strategies imposed by financialisation.

4.1.3 Financial channel: capital mobility and availability and policy space

Financialisation is a vehicle for more capital mobility and introducing new players and products into financial markets with objectives that are not always aligned with economic growth, stability or productive diversification. Financialisation practices increase uncertainty and the intensity and frequency of currency and financial crises in the periphery. Financial globalisation intensifies the contagion of global shocks in national, leaving less space for governments to conduct macroeconomic policies. In that way, it helps to cement the position of peripheral countries in the international monetary system and their place in the currency hierarchy, as they are seen as short-term places to invest with currencies that are ranked low and for whom it is difficult to climb due to this dynamic. In addition, all of these also has profound impacts on the productive structure. Instability, the lack of funds for productive investment or the shrinking in the policy space for conducting fiscal or industrial policies cement their position in the global productive system.

As we discussed in section 2, the international dimension of financialisation shows that cross-border capital flows are responsible for transmitting shocks worldwide, particularly in relation to property markets or domestic credit (Arestis and Glickman, 2002 and Akyuz, 2014). Movements in capital flows are more dependent on external conditions, typically on the conditions in central economies such as high liquidity and low-interest rates than on the fundamentals of peripheral economies (Andrade and Prates, 2013; Kalternbrunner and Painceira, 2017). This dynamic has been largely discussed in the literature (Bresser-Pereira, 2008; Ocampo, 2016; Paula et al., 2017; Fritz et al., 2018, Cimoli et al., 2020).

These dynamics cements the perception of these economies as highly unstable when comparing with the US or central economies. Investors do not invest long-term for fear of sudden changes. Domestic variables are subject to contagion because of the interconnectedness of the international financial system. Sustained periods of appreciation followed by depreciations have effects on competitiveness and on balance

sheets. The same is true for high interest rates. All these aspects reinforce the position of the countries in the global financial system and the pyramid of currencies, acting as a self-fulfilling prophecy. First, based on the risk perception, contracts of international investors are set mainly in US dollars; the same happens with import and export contracts, reinforcing the use of foreign currencies. Second, related to high levels of inflation or sudden movements on the exchange rate, for local agents, it is challenging to trust in their own currency as a store of value; therefore, several Latin American countries deposits are hoard in hard currencies, typically US dollars. Third, it is difficult to access external sources of credit in their own currency, so agents that have income in local currencies are indebted in US dollars ('original sin', Eichengreen and Hausmann, 1999); this had massive consequences derived from these currency mismatches, banking crises of Argentina and Uruguay in 2001-2002 are an example of it. Fourth, peripheral countries have higher external liabilities when comparing to external assets (Akyuz, 2020), generating an outflow of financial income from the periphery to the centres. Indicating that the gains from financialisation have not been equally distributed among both poles, enhancing the opportunities for the centres.

The effect that these conditions have on the periphery are significant. ECs have to offer high-interest rates to compensate for low liquidity and high-risk environments (Paula et al., 2017; Fritz et al., 2018). That impacts directly on investment as it is more expensive to finance production through indebtedness. In addition, capital inflows can generate exchange rate appreciations, affecting asset prices and real wages, with short periods of rises in investment, consumption and external debt. However, when there is a reversal in the cycle ('flight to quality'), capitals leave peripheral countries. Investors usually migrate to positions in currencies that offer greater liquidity and stability. The reversal affects not only the financial dimension but the real one. Deceleration of economic growth usually ends in economic crisis and significant depreciation, transforming production and consumption patterns and even generating changes in ownership structures (Fischer, 2015). In Latin America, the macroeconomic policies were not oriented as in Asia in controlling capital flows or pegging exchange rates to promote economic diversification and economic productivity; the prioritisation was on controlling inflation and macroeconomic equilibrium (Cimoli et al., 2020).

In the last years, derived from QE (Quantitative Easing) programs, the influx of capital bought assets, sovereign bonds and currencies (Bonizzi and Toporowski 2017). Central banks needed to engage in sterilisation strategies to control inflation, increasing the offer on public debt securities. These instruments were bought by banks and investment funds, expanding their balance sheets (Kalternbrunner and Painceira, 2017). Aggressive strategies towards household lending implied more household financialisation and a hike in asset prices; this dynamic resulted in more financial fragility (a la Minsky, 1978), potentially reinforcing the dynamic described paragraphs above. Central banks accumulated reserves to cope with changes in financial cycles and liquidity cycles in the centre. That is a typical macroeconomic accommodative policy in Structuralist terms. According to Lampa (2021), central banks treat capital flows as risky, the same as Prebisch did in the Central Bank of Argentina. This strategy came with high costs for ECs when comparing the remuneration paid on domestic instruments vs international reserves that do not receive or receive low remunerations (Rodrik, 2006)8. That resulted in fewer resources available for other purposes.

Financialisation, through the financial channel, the lack of resources available for productive investment affected the productive structure. The finance-led type of capitalism excluded finance for productive purposes, the financial sector benefited some high profitable sectors like construction and real estate, financially excluding firms (Favreau and Pérez Caldentey, 2019). In addition, development banks were debilitated as a result of neoliberal policies. Moreover, cuts in government spending to sustain 'sound' fiscal policies made fewer resources available for this type of activity. The lack of funds goes in hand with higher returns in the financial sector and its attractiveness versus real investment. In Latin America, Torija and Gottschalk (2018) found a rise in financial assets and high levels of leverage and an increase in the distribution of profits in Chile and Colombia. There is also evidence of national firms engaging in speculative activities (Demir, 2009; Rethel, 2010; Levy-Orlik, 2012; Correa et al., 2012; Powell, 2013) and carry trade operations (Kaltenbrunner and Panceira, 2017; and Bruno and Shin, 2015). NFCs from emerging countries have

[.]

⁸ Some scholars defend this strategy (see Bortz, 2018).

expanded their exposure to international financial markets, looking for investment opportunities abroad and taking debt in foreign currencies (Botta *et al.*, 2014).

Another implication of financialisation is related to the policy space available for developing countries to conduct economic growth policies and the upgrading of domestic industries (Chang, 2002; 2021). First, suppose ECs want to conduct active industrial policies to protect infant industries. In that case, they need foreign reserves, and countries rely on those reserves to control speculative attacks on currencies and preserve economic stability. The second one is related to the role of international institutions, the World Trade Organization, rating agencies, and the margin of manoeuvre they permit for a country to conduct their policies. Directly they are affected by multilateral and bilateral trade agreements (Wade, 2003) and indirectly by the possibility of rating agencies downgrading the qualification of government bonds if countries do not engage with 'sound' fiscal policies (Streeck, 2014).

To conclude this section, financialisation does not alleviate the centre-periphery division; on the contrary, it exacerbates it by affecting structural heterogeneity, imposing Balance of Payments constraints, enhancing commodity specialisation, or restricting the space available for industrial and fiscal reforms. Financialisation deepen economic volatility bringing more capital mobility where ECs cannot cope with those effects. All these aspects reinforce the position of the countries in the global financial system and in the pyramid of currencies.

4.2 Relation from the structural characteristics to financialisation

This section tries to disentangle how financialisation is characterised according to the structural features in the periphery. That means focusing on which actors are engaged in those practices, the reason for doing it, and the consequences. It presents how heterogeneity and the specialisation pattern shapes financialisation. Second, how economic cycles of capital and the terms of trade affect the periphery financialisation' style according to the place in the global financial and productive system.

4.2.1 Heterogeneity

As presented in Section 3, structural heterogeneity is a main characteristic of Latin American economies. It can be seen by the diverse levels of productivity among firms with different sizes and qualifications between workers. The periphery is heterogeneous because labour productivity is high in the exporting sector; meanwhile, the rest of the economic sectors are stagnant, and the economy specialises in few commodity goods. In the centre's case, it presents high levels of labour productivity across economic sectors and produces a wide range of goods with inter-sectorial complementation and integration across economic activities

In the periphery, small firms represent a massive proportion of the productive structure regarding employment, most firms present productivity levels below average and limited ability for innovation. The productive structure is simple, with some exceptional dynamic nucleolus usually related to GVCs (Global Value Chains) or state-owned firms. Most of the small and medium firms are disconnected from dynamic poles (Bértola 2015, pp. 279). In addition, most countries of Latin America present underdeveloped financial systems with low intermediation and are focused on short-term financing instruments. Firms encounter difficulties accessing financial services, even to the banking system (Manuelito and Jimenez, 2010). That is a stylized fact in the region, especially for small and medium-sized companies (see Stallings, 2006). According to the OECD (2013), small and medium-sized companies provide around 80% of jobs where only 15% of the credit is provided for this type of firm. Economic reforms did not come along with credit availability for many ECs; finance for productive purposes is scarce, especially for manufacturing (Naqvi, 2018; Favreau and Pérez Caldentey 2019).

In Latin America, only a small number of firms possess high productivity levels and can compete internationally, and those firms potentially have access to international capital markets (and domestic ones). Galindo and Schiantarelli (2003), found that for NFCs, access to finance depends not only on the balance sheet characteristics but also is more accessible for larger firms, foreign-owned or associated with a business group. Size is a variable also positively associated in Demir's analysis (Demir, 2009).

Presbitero and Rabellotti (2016) found a positive association between labour productivity and having better access to finance⁹. Access is still difficult, so we argue that financialisation has been **selective** within the Latin American context. In microeconomic terms, there is a particular type of firm prone to financialisation, those NFCs that are productive enough to operate in international markets. The orientation of financial deregulation schemes, and financialisation in general, has not been focused on enhancing productive firms across the board nor came accompanied by industrial or developmental plans.

The type of firm that can access financial markets is one that is good enough to compete internationally, NFCs that are large or productive enough to be part of it. Multinational Corporations (MNCs) and diversified business groups are the most important feature of the type of capitalism present in Latin America. Both types of firms are the responsible for mobilizing investment and are connected in a particular way with financialisation. For Roos Schneider (2013) Latin America has a type of capitalism that can be described as 'Hierarchical', where firms are controlled by wealthy families or foreign powers. The type of governance characterized by the relation between managers and family owners or with the headquarters of MCNs, is hierarchical, in relation to technology transfer, suppliers and customers relations. Business groups are a way of coping with economic volatility, as they can adapt fast by changing strategies, hiring, or firing workers faster than other firms. In addition to the ownership structure, where financial markets are not very deep, it is challenging to create a culture orientated to shareholder value (Torija and Gottschalk, 2018). High concentration levels lead to a more 'closed' system, where shares do not usually entitle the right to vote (for a detailed analysis, see Claessens and Yurtoglu, 2013 and OCDE, 2015). This is a main difference with rich countries' type of financialisation.

That does not imply that NFCs in ECs do not present signs of financialisation, there is evidence that some Latin American companies increased their amount of financial assets (Torija and Gottschalk, 2018); for example, engaging in carry trade operations (Kaltenbrunner and Panceira, 2017; and Bruno and Shin, 2015). The type of firm that can invest in profitable financial assets or use financial products to hedge against

⁹ In the case of Turkish firms, Akkemiek and Ozen, 2014, found a positive association between financialisation and the size of the firm.

volatility, possess an advantage over the rest, favouring a concentration process, thus reinforcing heterogeneity. Still, we need a better understanding of these firms, a characterisation of the economic sector they are in, the size, and the links with national and international firms to assess the impact on the productive structure. Nevertheless, we can argue that those firms that can take advantage of financialisation are those that, in a way, are the strongest within countries.

4.2.2 Specialisation pattern

The type of goods and services produced in the periphery, the specialisation pattern, relate national economies with the world economy. The role of two actors is key in maintaining the countries' position in the world division of labour, Transnational Corporations (TNCs) and national elites. We will present how the specialisation pattern cohabits, or creates, a particular type of financialisation.

TNCs have an essential role in intensifying labour division across the globe where the periphery focuses on natural resources or cheap labour. This aspect has been largely addressed by Structuralists and Dependentists (Furtado, Cardoso and Faletto, Prebisch, among others). Through GVC, there is intense competition among the lower value-added parts of the pyramid, with low wages and profit margins, typically located on the periphery. In the upper parts, innovation is produced, and finance and marketing activities are located. This type of division is based on controlling technology and property rights at the top (Medeiros and Trebat, 2016). Multilateral and bilateral agreements assure monopolistic positions for TNCs, so developing countries have difficulties escaping from the lower levels of international technological hierarchies. The position in the hierarchy is determinant for benefiting from innovation and being able to profit from it. According to Cimoli et al. (2005), a dominant position in the hierarchy guarantees "the control of knowledge decodification mechanisms" (p 34), determining the access to knowledge and the capacity to spread it across the rest of the productive structure. According to Fischer (2015), even Southeast Asian countries that incorporated more technological

capacities are trapped in subordinate positions in the international production structures.

Hierarchical capitalism has a competitive advantage derived from commodity production, related to natural conditions and a type of institutions that exploit these advantages (Roos Schneider, 2013). Also, MCNs and business groups did not demand considerable amounts of skilled workers nor investment in R&D, even when compared with business groups in other countries such as Korea or Sweden. That discouraged workers from investing in education or other abilities¹⁰. Financialisation is compatible with this pattern of insertion in the global productive system, that is why we called **selective** and **functional**. It did not help the periphery to upgrade technologically, on the contrary, through the neoliberal agenda facilitated corporate mobility and made it easier for TNCs to shift profits to low tax jurisdictions, increasing rents for these firms. TNCs have been pushing for the appearance and diffusion of GVCs, through the re-design of international rules to protect their extractive practices (Medeiros and Trebat, 2016; Andreoni and Chang, 2019). TNCs were established in the periphery to exploit the abundance of natural resources and cheap labour and expanded their profitability through financialisation. Global production strategies paved the way for sustained levels of financialisation (Milberg, 2008 pp. 445). Similar conclusions are found in Baud and Durand (2012) and Auvray and Rabinovich (2019). The initial global division of labour is reinforced by financialisation. First, by providing finance to their subsidiaries and at the same time relocating profits in offshore jurisdictions, making fewer funds available for national governments, acting as 'enclaves' within countries. Second, as it is more difficult for national firms to climb the technological ladder. Corporate mobility gave TNCs a competitive edge, making it more difficult for other companies to compete, particularly those with headquarters outside financial centres (De Medeiros 2017). In addition to business concentration and the ability for national companies to compete with TNCs, this regime meant fewer tax revenues available for national governments to conduct policies, influencing productive structures, for example, through less public investment in education or innovation.

-

¹⁰ Examples like Embrarer in Brazil, are more an exception than the rule in the continent

In the case of national firms, economic elites in the region are not willing to promote export-led industrialisation or a catching-up period in the periphery. As we mention, in order to industrialise is necessary a strong state's strategy to implement policies and manage conflicts among economic groups and convince domestic actors to put forward this agenda, but this is not easy to achieve if the domestic elite 'captured' the state (Pama, 2019). As Palma indicates a main difference between the industrialisation process in Asia and the Import Substitution Industrialisation (ISI) process in Latin America was the incapacity to establish mechanisms in which industrialisation could be sustained when the state support ended. The support for companies was not oriented, as in the infant industries, to be temporary until firms become internationally competitive (Fajnzylber, 1990). Some industrial development was possible in the continent during the ISI, accompanied by restrictions to capital movements and stable exchange rates with devaluated currencies for exports and appreciated for imports (Vernengo, 2005). However, most of it came to an end when the state support finished, and neoliberal policies were set in place.

Nowadays, several countries possess abundant natural resources and do not export raw commodities; they stimulate processing activities. That is the case of Finland, Sweden, Denmark, Malaysia or Vietnam. In the case of Latin America, 'there is not even an attempt to get out of this trap, preferring to follow the most comfortable path of exporting raw commodities' (Palma, 2019). Therefore, the problem is related to the lack of support for industrial policies, rather than with the abundance of natural resources. The roots of this behaviour can be traced back to the origins of these republics; some of the historical reasons for creating the modern democracies were to export raw materials out of the continent where social pacts were established to maintain the links with a transnational alliance. The neoliberal waves of the 70s and 90s are a new expression of this rooted dynamic (Deniz and Marshall, 2018). A successful strategy must combine industrial and fiscal policies but also manage the capital account and monetary policies. Otherwise, periods of real exchange rate appreciation and weak or industrial policies can end in the loss of technological capabilities affecting domestic production (Cimoli et al., 2020; Medeiros, 2020).

Domestic elites do not seem to be opposing financialisation and financial liberalisation. At least, financialisation did not interfere with the interest of national

commodity producers or TNCs, and it has been functional to their interests. In that way, financialisation could be seen as a 'new pact' between financial powers (domestic and international) and domestic elites related to commodity exploitation.

4.2.3 Economic cycles and terms of trade

Economic cycles are a characteristic of the capitalist system, where the periphery is subject to changes in liquidity and business cycles that are originated in the centres. Volatility is negative for long-term growth because it reduces investors' planning horizons and increases risk perceptions, discouraging productive investment and distorting macroeconomic prices. The increase in volatility is associated with the capital account and with changes in the terms of trade. Neo Structuralists have called this 'balance of payment dominance,' this is a type of regime in which external shocks (terms of trade or cycles of liquidity) dominate the macroeconomic dynamics in the short term (inflation, consumption, real wages, aggregate demand, and exchange rates). Examples of shocks related to access to external financing are the debt crisis of the 1980s, the Mexican crisis (1994), the Asian crisis (1997), the Argentinian crisis (2001) and the financial crisis (2008). Plus, movements in terms of trade, especially since the 2000s when the commodity boom in prices started, were incredibly significant. Titelman and Perez Caldentey (2015) showed that real economic fluctuations in Latin America and the Caribbean are less intense but more frequent than in other world regions. They also found a relation between the cycle and the tendency, so cycles have a lasting effect on the productive structure, despite the conjunction. In addition, the financial system brings more pro-cyclicality through the behaviour of credit, where the periods of contractions in credit are longer than those in GDP (also see ECLA 2004, 2012). In addition, there is less availability of funds in the public sector because of the decrease in tax revenues during the down-ward phases, reinforcing the decline (French-Davis, 2015). NFCs do not have enough funds to expand their productive capacity if they wanted to grow or reconvert. The long-term consequences of this greater real volatility are the meagre economic growth of the region compared to others in the world. Managing the cycles implies smoothing real and monetary variables.

During the 1990s, it was thought that controlling inflation and achieving fiscal equilibrium would be enough to obtain macroeconomic stability. This target was going to be met by imposing free exchanges rates and complete capital account openness. In consequence, foreign savings and national savings will finance large volumes of investment. With their votes of confidence, financial agents would restrain national administrations' proclivity to create fiscal and monetary imbalances. Financial markets would determine key variables like liquidity levels, interest rates, or equilibrium exchange rates. The strategy was relatively successful in terms of inflation, but as we saw, capital inflows generated appreciation of exchange rates, with negative impacts on the productive side, and on the stability of real economies.

Another essential element is the movement in the terms of trade. Erten (2011) presents the discussion of the decline in the terms of trade at a country level¹¹, analysing for the period 1960-2006 trade relations between North and South. Concluding that terms of trade have been detrimental for the South since the 1960s, showing that the impact was more significant for exporters of manufactures of developing countries. In addition, in the last decade, commodity prices showed significant levels of volatility and sometimes a boom-and-bust trajectory (Akyüz, 2012). These movements lead firms to hedge themselves, buying financial products like forwards on the prices of commodities. In a way, financialisation practices are a consequence of this structural trend. Besides, firms have been speculating and increasing financial gains; this could be a strategy to take advantage of good times and to save for moments of declining prices.

These structural dynamics affected the type of financialisation. It has been **selective**, focused on short-term products and some economic agents. International investors can take advantage or easily escape from cyclicality; the same is true for some national investors, who can potentially take advantage or protect from the well-known volatility. Cyclicality affects producers disparately; therefore, instability is not neutral. Motives behind financialisation in the periphery relate to the fact that Latin American economies are volatile and have low currencies in the international hierarchy. The

-

¹¹ Erten (2011) summarises the empirical literature on the declining terms of trade with a discussion of the different econometric techniques used in each one.

style is dominated by financial investors (or NFCs that act as financial investors) that are short-term oriented and are used to take advantage of sudden changes. Therefore, financialisation has been **functional** to financial investors, who are not focused on long-term decisions. Financial investors can adapt more quickly to sudden changes in relative prices or expectations than industrial capitalists. Productive capitalists need longer periods to make investment or employment decisions as they are usually difficult to revert. At the same time, industries related to commodities can cope better with economic volatility than domestically oriented industries (Cimoli et al., 2020). Financialisation in the periphery is also related to precautionary motives; NFCs use financial products to protect themselves against volatility (Demir, 2009; Rossi, 2013; Kaltenbrunner and Panceira, 2017). There is also a trend for NFCs in ECs to hold high cash and liquid assets (Kalinowski and Cho, 2009; Karwowski, 2012).

5. Final remarks

Financialisation is a global phenomenon with different implications for the centre and the periphery; it requires to be analysed jointly. The integration of peripheral countries in the world economy needs to consider the financial dimension and the productive one. Economic relations between both poles tend to reproduce the conditions of underdevelopment in the periphery and increase the distance between them. Under this scheme, we analysed how countries' structural characteristics are affected by financialisation, finding that financialisation strengthens the subordinate position in the global productive and financial system. Financialisation deepens the centreperiphery dichotomy, increases structural heterogeneity, and amplifies business and financial cycles reinforcing the place in the global financial system and currency hierarchy. In the productive dimension by i) decreasing real investment, productivity, and expenditures in R&D, ii) reducing the policy space for industrial reforms, and iii) amplifying commodity boom-bust trajectories and re-primarization strategies, and from the financial side by i) increasing instability and recurrent economic crisis and the ii) lack of resources for long-term investment.

Second, we analysed how the structural characteristic of Latin American economies matters for financialisation phenomena. Financialisation in the periphery is selective and functional. On the internal front, it has been selective, as it is focalises

in some firms which are prone to financialisation; they can access financial markets to fund their activities, hedge themselves and obtain some gains from speculative activities. The rest of the economy is characterised by low levels of intermediation. It is characterised as functional as it did not alter the internal economic and political balances. It did not promote industrial and value-added activities that could potentially have conflicting interests with primary activities (regarding capital controls, exchange rate controls or actively infant industry protection). Regarding the external front, it has been selective as it is characterised by prominent financial integration levels, a short-term type of instruments that look for capital gains in the periphery and was not oriented to provide capital for improving socio-economic conditions or promoting a catching-up transition. It is functional as financialisation strengthened the position of these countries in the world order based on the original international division of labour, unaltering the characteristic of being countries specialised in commodity production and low value-added activities when comparing with central countries. That does not mean that it is impossible to catch up, as the history of some Asian countries can demonstrate, but it is necessary to conduct active policies. The style of financialisation in the periphery relates to the fact that Latin American economies are volatile due to their productive structure and capital account openness with currencies in the lower part of the monetary hierarchy.

References

Aalbers, M. B. 2008. The Financialization of Home and the Mortgage Market Crisis, *Competition and Change*, 12(2), pp. 148–166.

Aalbers, M. 2017. Corporate Financialization, *The International Encyclopedia of Geography*, pp. 1–10.

Abeles, M.; Pérez Caldentey, E. and Valdecantos, S. 2018 (ed), *Estudios sobre nancierización en América Latina*, Libros de la CEPAL, N° 152 (LC/PUB.2018/3-P), Santiago, Comisión Económica para América Latina y el Caribe (CEPAL).

Adrian, T., Ashcraft, A. and Pozsar, Z. 2010. Shadow Banking. Federal Reserve Bank of New York Staff Reports No.458

Adrian, T. and Shin, H. S. 2010. The Changing Nature of Financial Intermediation and the Financial Crisis of 2007-09, Federal Reserve Bank of New York Staff Reports. No 439

Aglietta, M and Breton, R. 2001. Financial systems, corporate control and capital accumulation, *Economy and Society*, 30(4), pp. 433-466

Akkemik, K. and Ozen, S. 2014. Macroeconomic and institutional determinants of financialisation of non-financial firms: Case study of Turkey, *Socio-Economic Review*, 12, pp. 71–98

Akyüz, Y. 2012. The Boom in Capital Flows to Developing Countries: Will It Go Bust Again? *ekonomi-tek international economics* 1(1), pp. 63–96.

Akyüz, Y. 2014. Internationalization of Finance and Changing Vulnerabilities in Emerging and Developing Economies. Discussion Paper No. 217. Geneva: UNCTAD.

Amarante, V. y J. P. Jiménez. 2015. Desigualdad, concentración y rentas altas en América Latina, *Desigualdad, concentración del ingreso y tributación sobre las altas rentas en América Latina*, Libros de la CEPAL, No 134 (LC/G.2638-P), Santiago, Comisión Económica para América Latina y el Caribe (CEPAL).

Andrade, R. P. and Prates, D. M. 2013. Exchange rate dynamics in a peripheral monetary economy', *Journal of Post Keynesian Economics*, 35(3), pp. 399–416

Araújo, E.; M. Bruno; and D. Pimentel. 2012. Financialization Against Industrialization: A Regulationist Approach of the Brazilian Paradox. *Revue de la régulation: Capitalisme*, institutions, pouvoirs, vol. 11.

Arellano, M., Bond, S. 1991. Some tests of specification for panel data: Monte Carlo evidence and an application to employment equations. *Review of Economic Studies*, (58), pp. 277–297.

Arestis, P. and Glickman, M. 2002. Financial Crisis in South East Asia: dispelling illusion the Minskyan Way, *Cambridge Journal of Economics*, (26), pp. 237–260.

Arocena, R. and Sutz, J.2000. Looking at National System of Innovation from the South. *Industry and Innovation*, 7(1), pp. 55-75

Ashman, S.; B. Fine; and S. Newman. 2011. The Crisis in South Africa: Neoliberalism, Financialization and Uneven and Combined Development. *Socialist register* 47, pp. 174–95.

Becker, J., Jäger, J., Leubolt, B. & R. Weissenbacher. (2010) 'Peripheral Financialization and Vulnerability to Crisis: A Regulationist Perspective', *Competition & Change*, 14(3), pp. 225–247.

Bértola, L. 2015. Patrones de desarrollo y Estados de bienestar en América Latina.. In: Alicia Bárcena, Antonio Prado (eds.) *Neoestructuralismo y Corrientes Heterodoxas em América Latina y el Caribe a Inicios del Siglo XXI*, Santiago do Chile, CEPAL, IDRC, CRDI.

Bértola, L. and Ocampo. J.A. 2010. Desarrollo, vaivenes y desigualdad: Una historia económica de América Latina desde la independencia, Madrid, Secretaría General Iberoamericana (SEGIB).

Bielschovsky, R. 1998. Evolución de las ideas de la CEPAL, *Revista CEPAL*, No extraordinario (LC/G.2037-P/E), Santiago de Chile, Comisión Económica para América Latina y el Caribe (CEPAL).

BIS (2007) Evolucion de los sistemas bancarios en América Latina y el Caribe : retos e implicaciones para la política monetaria y la estabilidad financiera. BIS Papers. No 33.

Bonizzi, B. (2013) 'Financialization in Developing and Emerging Countries', *International* 55 *Journal of Political Economy*, 42(4), pp. 83–107.

Bonizzi, B. 2017. International Financialisation, Developing Countries and the Contradictions of Privatised Keynesianism, *Economic and Political Studies* 5(1), pp. 21–40

Bortz, P. G. and Kaltenbrunner, A. 2017. The International Dimension of Financialization in Developing and Emerging Economies, *Development and Change*, 49(2), pp. 375-393

Bortz, P.G. 2018. Flujos de Capital y Endeudamiento Externo: Algunas Reflexiones para el Caso Latinoamericano ['Capital Flows and External Indebtedness: Some Reflections on the Latin American Case'], in M. Abeles and S. Valdecantos (eds) *Estudios sobre Financiarización en América Latina* [Studies on Financialization in Latin America]. Santiago: CEPAL.

Botta, A., A. Godin and M. Missaglia. 2014. 'Finance, Foreign (Direct) Investment and Dutch Disease: The Case of Colombia'. DEM Working Paper Series No. 90 (09-14). Pavia: University of Pavia.

Boyer, R. 2000. Is a finance-led growth regime a viable alternative to Fordism? A preliminary analysis. *Economy and Society*, *29*(1), pp. 111–145.

Bruno, V. and Shin H. S. 2017, Global dollar credit and carry trades: a firm-level analysis;, *Review of Financial Studies*, 30(3), pp. 703–749.

CAF, 2011. Servicios financieros para el desarrollo: Promoviendo el acceso en América Latina. Bogotá: Corporación Andina de Fomento.

Cimoli, M. and others. 2005. Cambio estructural, heterogeneidad productiva y tecnología en América Latina", *Heterogeneidad estructural, asimetrías tecnológicas y crecimiento en América Latina*, M. Cimoli (ed.), Santiago de Chile, Comisión Económica para América Latina y el Caribe (CEPAL)/Banco Interamericano de Desarrollo (BID).

Cimoli, M; Ocampo, J; Porcile, G and Saporito, N. 2020. Choosing sides in the trilemma: international financial cycles and structural change in developing economies, *Economics of Innovation and New Technology*, 29(7), pp. 740-761

Chena, P.; Buccella, M. and Bosnic, C. 2018. Efectos de la financierización en el cambio tecnológico en América Latina in M. Abeles and S. Valdecantos (Eds) *Estudios sobre Financiarización en América Latina* [Studies on Financialization in Latin America]. Santiago: CEPAL.

Chinn, M. and Ito, H. 2006. What matters for financial development? Capital controls, institutions, and interactions. *Journal of Development Economics*, 81, pp. 163–192.

Christophers, B. 2012. Anaemic Geographies of Financialisation, *New Political Economy*, 17(3), pp. 271-291.

Christophers, B. 2015. The limits to financialisation, *Dialogues in Human Geography*, 5(2), pp. 183–200

Claessens, S. y B. Yurtoglu. 2013, Corporate governance in emerging markets: a survey, *Emerging Markets Review*, vol. 15, pp.1-33.

Correa, E. 2006. Banca extranjera en America Latina, in CLACSO (ed.) *Reforma Financiera en America Latina*. Buenos Aires, pp. 67–81.

Correa, E., Vidal, G. and Marshall, W. 2012. Financialization in Mexico: trajectory and limits, *Journal of Post Keynesian Economics*, 35(2), pp. 255–276.

Crotty, J. 2003. The Neoliberal Paradox: The Impact of Destructive Product Market Competition and Impatient Finance on Nonfinancial Corporations in the Neoliberal Era, *Review of Radical Political Economics*, 35(3), pp. 271–279.

Demir, F. 2007. The Rise of Rentier Capitalism and the Financialization of Real Sectors in Developing Countries, *Review of Radical Political Economics*, 39(3), pp. 351–359.

Demir, F. 2009. Financial liberalization, private investment and portfolio choice: Financialization of real sectors in emerging markets, *Journal of Development Economics*, 88(2), pp. 314–324.

Dymski, G.A. 1999. Asset Bubbles and Minsky Crises in East Asia: A Spatialized Minsky Approach. Research paper, Department of Economics, University of California-Riverside.

Dosi, G., Revest, V. and Sapio, A. 2016. Financial regimes, financialization patterns, and industrial performances: Preliminary remarks. *Revue d'économie industrielle* 154 (2), pp 63–96.

Duménil, G., and Lévy, D. 2004. Capital resurgent: Roots of the neoliberal revolution. Harvard University Press.

Engelen, E. 2008. The Case for Financialization, *Competition and Change*, 12, pp. 111–119.

Epstein, G.A. 2005. Introduction: Financialization and the World Economy. In *Financialization and the world economy*, ed. G.A. Epstein, 3–16. Cheltenham: Edward Elgar.

Epstein, G. 2015. *Financialization: there's something happening here*. Political Economy Research Institute Working Paper #394.

Erten, B. 2011. North-South Terms-of-Trade Trends from 1960 to 2006, *International Review of Applied Economics* 25(2): 171–84.

Erturk, I. and Solari. S. 2007. Banks as Continuous Reinvention, *New Political Economy*, 12(3), pp. 369–88.

Favreau, N. and Pérez Caldentey, E. 2019. *Financialisation in a small, open developing economy: the case of Chile.* In Levy, N. and Bustamante, J. (2019)(editors) Financialisation in Latin America. Challenges of the export-led growth model. Routledge, London.

Fischer, A. M. (2015) 'The end of peripheries? On the enduring relevance of structuralism for understanding contemporary global development', *Development and Change*, 46(4), pp. 700–732.

Fligstein, N. and Goldstein, A. 2015. The emergence of a finance culture in American households, 1989-2007, *Socio-Economic Review*, 13(3), pp. 575–601.

Gabor, D. 2010. Central Banking and Financialization. A Romanian Account of how Eastern Europe became Subprime, London, Palgrave Macmillan.

Gabor, D. V. 2012. The Road to Financialization in Central and Eastern Europe: The Early Policies and Politics of Stabilizing Transition, *Review of Political Economy*, 24(2), pp. 227–249.

Hausmann, R. 2011. Structural transformation and economic growth in Latin America, *The Oxford Handbook of Latin American Economics*, J.A. Ocampo and J. Ros, Oxford University Press.

Hein, E. 2011. Distribution, Financialisation' and the Financial and Economic Crisis - Implications for Post-crisis Economic Policies. MPRA Paper No.31180. Available at: http://mpra.ub.uni-muenchen.de/31180/.

Hein, E. 2013. The crisis of finance-dominated capitalism in the euro area, deficiencies in the economic policy architecture, and deflationary stagnation Policies, *Journal of Post Keynesian Economics*, 36(2), pp. 325–354.

Hein, E. 2015. Finance-dominated capitalism and re-distribution of income: A Kaleckian perspective, *Cambridge Journal of Economics*, 39(3), pp. 907–934.

IMF. 2015. Global Financial Stability Report, October 2015. Vulnerabilities, Legacies, and Policy Challenges: Risks Rotating to Emerging Markets. Washington, DC: International Monetary Fund.

Jibril, H.; Kaltenbrunner, A.; and Kesidou, E. 2018. Financialisation and Innovation in Emerging Economies: Evidence from Brazil. Leeds University Business School Working Paper No.

18-09, Available at SSRN: https://ssrn.com/abstract=3188021 or http://dx.doi.org/10.2139/ssrn.3188021

Kalinowski, T., and H. Cho. 2009. The Political Economy of Financial Liberalizatio in South Korea: State, Big Business, and Foreign Investors. *Asian Survey* 49(2), pp. 221–42.

Kaltenbrunner, A. and J.P. Painceira. 2015. Developing Countries' Changing Nature of Financial Integration and New Forms of External Vulnerability: The Brazilian Experience, *Cambridge Journal of Economics* 39(5), pp. 1281–306.

Kaltenbrunner, A. and J.P. Painceira. 2017. Subordinated Financial Integration and Financialisation in Emerging Capitalist Economies: The Brazilian Experience, *New Political Economy*. 23(3), pp. 290-313

Karacimen, E. 2016. Consumer credit as an aspect of everyday life of workers in developing countries. Evidence from Turkey, *Review of Radical Political Economics*, vol. 48, No 2, Thousand Oaks, SAGE Publishing.

Karwowski, E. 2012. Financial Operations of South African Listed Firms: Growth and Financial Stability in an Emerging Market Setting. Paper presented at *III Conferencia international do IESE*, Mozambique September 3–4.

Karwowski, E. and Stockhammer, E. 2017. Financialisation in emerging economies: a systematic overview and comparison with Anglo-Saxon economies, *Economic and Political Studies*, 5(1), pp. 60-86.

Karwowski, E.; Shabani, M. and Stockhammer, M. 2020. Dimensions and Determinants of Financialisation: Comparing OECD Countries since 1997, *New Political Economy*, 25(6), pp. 957-977.

Katz, J. 2020. Crecer a partir de recursos naturales. Aspectos macro y microeconómicos, temas regulatorios y el nuevo debate sobre "derechos ambientales". Teoría y políticas públicas in *Teoría de la innovación: evolución, tendencias y desafíos : herramientas conceptuales para la enseñanza y el aprendizaje* / Diana Suárez... [et al.]: Universidad Nacional de General Sarmiento; Madrid: Ediciones Complutense, 2020. Libro digital, PDF

Krippner, G. 2005. The Financialization of the American Economy. *Socio-Economic Review*, 3(2), pp. 173–208.

Langley, P. 2008. Financialization and the Consumer Credit Boom, *Competition and* 57 *Change*, 12(2), pp. 133–147.

Lapavitsas, C. 2009a. Financialisation Embroils Developing Countries, *Papeles de Europa*, 19, pp. 108–139. Available at: http://revistas.ucm.es/portal/abrir.php?url=http://revistas.ucm.es/cee/19895917/articulos/PADE0909220108A.PDF.

Lapavitsas, C. 2009b. Financialised Capitalism: Crisis and Financial Expropriation, *Historical Materialism*, 17(2), pp. 114–148.

Lapavitsas, C. and Powell, J. 2013. Financialisation varied: A comparative analysis of advanced economies, *Cambridge Journal of Regions, Economy and Society*, 6(3), pp. 359–379.

Lapavitsas, C. 2014. Profiting Without Producing How Finance Exploits Us All. London: Verso.

Lazonick, W. and O'Sullivan, M. 2000. Maximizing shareholder value: a new ideology for corporate governance, *Economy and Society*, 29(1), pp. 13–35.

Lazonick, W. and Mazzucato, M. 2013. The risk-reward nexus in the innovation-inequality

relationship: who takes the risks? Who gets the rewards?. *Industrial and Corporate Change*, 22(4), pp. 1093-1128.

Lazonick, W. 2014. The Big Idea: profits without prosperity, *Harvard Business Review*, Setiembre, pp. 47–55.

Lee, Y.; Kim, H. and Hwan Joo, S. 2020. Financialisation and Innovation Short-termism in OECD Countries. *The Review of radical political economics*, 52 (2), pp. 259-286.

Levine, R. 1997. Financial development and economic growth: views and agenda. *Journal of Economic Literature*, 35, pp. 688–72.

Levine, R. and King, R. G. 1993. Finance, entrepreneurship and growth: Theory and evidence, *Journal of Monetary Economics*, 32(1993), pp. 513–542.

Levy-Orlik, N. 2012. Effects of Financialization on the Structure of Production and Non-financial Private Enterprises: The Case of Mexico, *Journal of Post Keynesian Economics*, 35(2), pp 235–54.

Manuelito, S. and Jimenez, L. F. 2010. Los mercados Financieros en América Latina, Macroeconomía para el desarrollo. No 107. CEPAL, Santiago de Chile. Available at: http://repositorio.cepal.org/bitstream/handle/11362/5337/lcl3270.pdf?sequence=1.

Medeiros, C. 2008 Financial dependency and growth cycles in Latin American countries. *Journal of Post Keynesian Economics*. 31(1), pp. 79-99.

Medeiros, C., and Trebat, N. 2016. Finance, trade and income distribution in global value chains: Implications for developing economies and Latin America. Retrieved from http://www.ie.ufrj.br/images/pesquisa/publicacoes/discussao/2017 /tdie0022017me deirostrebat.pdf

Michell, M and Toporowski, J. 2013. Critical Observations on Financialization and the Financial Process, *International Journal of Political Economy*, 42(4), pp.67-82

Ocampo, J. A and Parra M. 2003. The terms of trade *for commodities* in the twentieth *century*. *Revista CEPAL*, (LC/G.2200-P) Santiago de Chile, Comisión Económica para América Latina y el Caribe (CEPAL).

Ocampo, J.A., C. Rada y L. Taylor. 2009, *Growth and Policy in Developing Countries: A Structuralist Approach*, Oxford University Press

Ocampo, J.A. 2011. Macroeconomía para el desarrollo: políticas anticíclicas y transformación productiva, *Revista CEPAL*, No 104 (LC/G.2498-P/E), Santiago de Chile, Comisión Económica para América Latina y el Caribe (CEPAL).

Ocampo, J.A. 2014. Latin American structuralism and production development strategies. ILO, Geneva.

Ocampo, J. A. 2015. America Latina frente la Turbulencia Economica Mundial. In: Alicia Bárcena, Antonio Prado (eds.) *Neoestructuralismo y Corrientes Heterodoxas em América Latina y el Caribe a Inicios del Siglo XXI*, Santiago de Chile, CEPAL, IDRC, CRDI.

OCDE. 2015. Corporate Governance of Company Groups in Latin America, París.

OIT/OCDE. 2015. *The labour share in G20 economies*, report for G20 group, available at https://www.oecd.org/g20/topics/employment-and-social-policy/The-Labour-Share-in-G20-Economies.pdf.

Onaran, Ö. and Tori, D. 2016. The effects of financialization on investment: Evidence from firm-level data for the UK. Post Keynesian Economics Study Group Working Paper. PKSG *No* 1601.

Orhangazi, O. 2008. Financialisation and capital accumulation in the corporate sector: 1973-2003, *Cambridge Journal of Economics*, 32(6), pp. 863–886.

Orozco, J. 2020. Una visión crítica del concepto de transferencia tecnológica y de conocimiento. in *Teoría de la innovación: evolución, tendencias y desafíos : herramientas conceptuales para la enseñanza y el aprendizaje* / Diana Suárez... [et al.] Universidad Nacional de General Sarmiento ; Madrid : Ediciones Complutense, 2020. Libro digital, PDF

Painceira, J. P. 2009. Developing Countries in the Era of Financialisation: From Deficit Accumulation to Reserve Accumulation, *Research on money and finance Discussion Papers*. *No* 4.

Painceira, J. P. 2011. Central Banking in Middle Income Countries in the Course of Financialisation: A Study with Special Reference to Brazil and Korea. PhD Dissertation, SOAS, University of London.

Palma, G. 1998. Three and a Half Cycles of 'Mania, Panic, and [Asymmetric] Crash': East Asia and Latin America Compared. *Cambridge Journal of Economics*, 22(6) pp. 789–808.

Park, J; Lee, B; Moon, Y; Kim, G; Kwon, L. 2018. Relation of R&D expense to turnover and number of listed companies in all industrial fields, *Journal of Open Innovation: Technology, Market, and Complexity* 4(9), pp. 1-15

de Paula, L.; Fritz, B. and Prates, D. (2017) 'Keynes at the periphery: Currency hierarchy and challenges for economic policy in emerging economies', *Journal of Postkeynesian Economics*, 40(2), pp 183-202

Pérez Caldentey, E. 2015. Una coyuntura propicia para re exionar sobre los espacios para el debate y el diálogo entre el (neo)estructuralismo y las corrientes heterodoxas. In: Alicia Bárcena, Antonio Prado (eds.) *Neoestructuralismo y Corrientes Heterodoxas em América Latina y el Caribe a Inicios del Siglo XXI*, Santiago de Chile, CEPAL, IDRC, CRDI.

Pike, A. y J. Pollard. 2010. Economic geographies of financialization, *Economic Geography*, vol. 86, No 1, Hoboken, Wiley

Powell, J. 2013. Subordinate financialisation: A study of Mexico and its non-financial corporations. PhD Thesis. SOAS, University of London.

Prebisch, R., 1949. *The Economic Development of Latin America and Some of Its Problems*. ECLA, New York.

Rethel, L. 2010. Financialisation and the Malaysian Political Economy, *Globalizations*, 7(4): pp. 489–506.

Robertson, M. 2016. (De)constructing the financialised culture of owner-occupation in the UK, with the aid of the 10Cs, *New Political Economy*. Taylor & Francis, 22(4), pp. 398–409.

Rodriguez, O. 1980. La teoría del subdesarrollo de la CEPAL: síntesis y crítica. *Revista Comercio Exterior*, 30(12), pp. 1346-1362.

Sanchis-arellano, A. and Rinaldi, L. 2006. *Household Debt Sustainability What Explains Household Non-performing Loans?*. European Central Bank. Working Paper Series. No 570.

Sawyer, M., 2013–14. What Is Financialization?. *International Journal of Political Economy*, 42(4), pp. 5–18

dos Santos, P. 2009. On the Content of Banking in Contemporary Capitalism, *Historical Materialism* 17(2), pp. 180–213.

dos Santos, P. L. 2013. A Cause for Policy Concern: The Expansion of Household Credit in Middle-Income Economies, *International Review of Applied Economics*, 27(3), pp. 316–338.

Seo, H.J., H.S. Kim y Y.C. Kim 2012. Financialization and the slowdown in Korean firms. R&D investment, *Asian Economic Papers*, 11(3), Cambridge, The MIT Press.

Sokol, M. 2017. Financialisation, financial chains and uneven geographical development: towards a research agenda, *Research in International Business and Finance*, vol. 39, Amsterdam, Elsevier.

Stockhammer, E. 2004. Financialisation and the slowdown of accumulation, *Cambridge Journal of Economics*, 28(5), pp. 719–741.

Stockhammer, E. 2008. Some Stylized Facts on the Finance-dominated Accumulation Regime', *Competition & Change*, 12(2), pp. 184–202.

Stockhammer, E. 2010. Financialization and the Global Economy. Political Economy Research Institute Working Paper No. 242. Amherst, MA: University of Massachusetts, Amherst.

Stockhammer, E. 2012. Financialization, income distribution and the crisis, *Investigación Económica*, LXXI(279), pp. 39-70.

Stockhammer, E. 2013. Why have wage shares fallen? A panel analysis of the determinants of functional income distribution, *Conditions of Work and Employment Series*, No 35, Ginebra, Organización Internacional del Trabajo (OIT).

Streeck, W. 2017. Buying Time: The Delayed Crisis of Democratic Capitalism. Second, ITTO Tropical Forest Update. Second. London: Verso.

Tang, K., and Xiong, W. 2010 Index Investment and Financialization of Commodities. Working Paper no. 16385, *National Bureau of Economic Research*, Cambridge, MA.

Tavares, M. C. and Gomes, G. 1998. La CEPAL y la integración económica de América Latina. Revista de la CEPAL, número extraordinario, CEPAL, Santiago de Chile.

Torija, E., and Gottschalk, R. 2018. Patrones financieros y de inversión en América Latina

desde la perspectiva del comportamiento empresarial. In M. Abeles, E. Pérez Caldentey, & S. Valdecantos (Eds.), *Estudios sobre financierización en América Latina*. Santiago: CEPAL.

UNCTAD 2014. World Investment Report. Geneva: UNCTAD.

UNCTAD and FAO. 2017. Commodities and Development Report 2017 Commodity Markets, Economic Growth and Development, New York.

Van der Zwan, N. 2014. Making sense of financialization. *Socio-Economic Review*, 12(1), pp. 99–129.

Vernengo, M., 2005. Technology, finance, and dependency: Latin American radical political economy in retrospect. *Review of Radical Political Economics*, 38(4), pp.551-568.

Wade, R. 2018. The developmental state: dead or alive?, Development and Change, 49, pp 518–546.

